Author's response to reviews

Title: Pride and confidence at work: potential predictors of occupational health in a hospital setting

Authors:

Kerstin Nilsson (kerstin.nilsson@his.se)
Anna Hertting (anna.hertting@ipm.ki.se)
Inga-Lill Petterson (ingga-lill.pettersson@sll.se)
Tores Theorell (tores.theorell@ipm.ki.se)

Version: 4 Date: 26 August 2005

Author's response to reviews: see over
Gothenburg, 26/08/2005

The Biomed Central Editorial Team

Title: Pride and confidence at work: potential predictors of occupational health in a hospital setting

Author: Kerstin Nilsson, Anna Hertting, Inga-Lill Petterson, Töres Theorell

Thank you for accepting the manuscript above. The following changes are made in the article:

1. The title has been formatted with sentence case.
2. All highlighting/tracking from the previous version is removed.
3. The figure and table placement text are removed.
4. Abstract: The words manageable, comprehensive and meaningful in the result section are changed to manageability, comprehensiveness and meaningfulness.
5. Method, Participants and data collection: t in (see table 1) is changed to T.
6. Result: The word, see, is inserted before Table in (Table 2).
7. Result, Supportive activities: In quotation 16 head of the department is changed to head of the ward. This change is also made in the text above the quotation.
8. Conclusion: The words manageable, comprehensive and meaningful are changed to manageability, comprehensiveness and meaningfulness.
9. References: Translation of the Swedish title of reference 20 is changed to “Sustainable health at work in municipalities and county councils”.

Sincerely yours
Kerstin Nilsson on behalf of the authors